User's Manual

AQ6370 Viewer



IM 735371-01E 11th Edition

	Thank you for purchasing the AQ6370 Viewer. This user's manual lists precautions that must be taken during use, and contains useful information about the functions and operating procedures of the software program. To ensure correct use, please read this manual thoroughly during operation. Keep this manual in a safe place for quick reference in the event a question arises. In addition to this manual, the AQ6370 Viewer Installation Manual (IM 735371-02E) is provided for installing and running the AQ6370 Viewer. For information about the handling precautions, functions, and operating procedures of the Optical Spectrum Analyzer (AQ6370 series) and the handling and operating procedures for Windows, see the manuals for those products.
Notes	 The contents of this manual are subject to change without prior notice as a result of continuing improvements to the instrument's performance and functions. The figures given in this manual may differ from those that actually appear on your screen. Every effort has been made in the preparation of this manual to ensure the accuracy of its contents. However, should you have any questions or find any errors, please contact your nearest YOKOGAWA dealer. Copying or reproducing all or any part of the contents of this manual without YOKOGAWA's permission is strictly prohibited.
Trademarks	 Microsoft, Microsoft Internet Explorer, MS-DOS, and Windows are either registered trademarks or trademarks of Microsoft Corporation in the United States and/or other countries. Adobe, Acrobat, and PostScript are trademarks of Adobe Systems Incorporated. For purposes of this manual, the
Revisions	 1st Edition October 2006 2nd Edition May 2008 3rd Edition October 2008 4th Edition February 2010 5th Edition October 2010 6th Edition May 2014 7th Edition March 2015 8th Edition March 2016 9th Edition March 2017 10th Edition May 2018 11th Edition August 2020

Terms and Conditions of the Software License

Yokogawa Electric Corporation and Yokogawa Test & Measurement Corporation, Japanese corporations (hereinafter called "Yokogawa"), grants permission to use this Yokogawa Software Program (hereinafter called the "Licensed Software") to the Licensee on the conditions that the Licensee agrees to the terms and conditions stipulated in Article 1 hereof.

You, as the Licensee (hereinafter called "Licensee"), shall agree to the following terms and conditions for the software license (hereinafter called the "Agreement") based on the use intended for the Licensed Software.

Please note that Yokogawa grants the Licensee permission to use the Licensed Software under the terms and conditions herein and in no event shall Yokogawa intend to sell or transfer the Licensed Software to the Licensee. Licensed Software Name: USB Compliance Test Software (Model 701985/F30)

Number of Client License: 1

Article 1 (Scope Covered by these Terms and Conditions)

- 1.The terms and conditions stipulated herein shall be applied to any Licensee who purchases the Licensed Software on the condition that the Licensee consents to agree to the terms and conditions stipulated herein.
- 1.2The "Licensed Software" herein shall mean and include all applicable programs and documentation, without limitation, all proprietary technology, algorithms, and knowhow such as a factor, invariant or process contained therein.

Article 2 (Grant of License)

- 2.1Y0kogawa grants the Licensee, for the purpose of single use, non-exclusive and non-transferable license of the Licensed Software with the license fee separately agreed upon by both parties. Yokogawa allows the Licensee to install only one instance of the software for use on one device, whether that device is physical or virtual. If the Licensee want to use the software on more than one virtual device, the Licensee must obtain a separate Licensed Software for each instance.
- 2.3The Licensee is, unless otherwise agreed in writing by Yokogawa, not entitled to copy, change, sell, distribute, transfer, or sublicense the Licensed Software. 2.3The Licensee Software shall not be copied in whole or in part except for keeping one (1) copy for back-up purposes. The Licensee shall secure or supervise the copy of
- the Licensed Software by the Licensee itself with great, strict, and due care. 2.4In no event shall the Licensee dump, reverse assemble, reverse compile, or reverse engineer the Licensed Software so that the Licensee may translate the Licensed Software into other programs or change it into a man-readable form from the source code of the Licensed Software. Unless otherwise separately agreed by Yokogawa, Yokogawa shall not provide the Licensee the source code for the Licensed Software.
- 2.5The Licensed Software and its related documentation shall be the proprietary property or trade secret of Yokogawa or a third party which grants Yokogawa the rights. In no event shall the Licensee be transferred, leased, sublicensed, or assigned any rights relating to the Licensed Software.

- 2.6 Yokogawa may use or add copy protection in or onto the Licensed Software. In no event shall the Licensee remove or attempt to remove such copy protection.
 2.7 The Licensed Software may include a software program licensed for re-use by a third party (hereinafter called "Third Party Software", which may include any software program from affiliates of Yokogawa made or coded by themselves.) In the case that Yokogawa is granted permission to sublicensee to third parties by any licensors (sub-licensor) of the Third Party Software pursuant to different terms and conditions than those stipulated in this Agreement, the Licensee shall observe such terms and conditions of which Yokogawa notifies the Licensee in writing separately. 2.8Yokogawa software products may contain open source software (hereafter referred to as OSS). For the OSS, terms and conditions provided separately apply with
- higher precedence than those of this agreement.

Article 3 (Restriction of Specific Use)

3.1The Licensed Software shall not be intended specifically to be designed, developed, constructed, manufactured, distributed or maintained for the purpose of the following events:

- Operation of any aviation, vessel, or support of those operations from the ground;, a)
- b)
- Operation of nuclear products and/or facilities;, Operation of nuclear weapons and/or chemical weapons and/or biological weapons; or
- d) Operation of medical instrumentation directly utilized for humankind or the human body.
 3.2Even if the Licensee uses the Licensed Software for the purposes in the preceding Paragraph 3.1, Yokogawa has no liability to or responsibility for any demand or damage arising out of the use or operations of the Licensed Software, and the Licensee agrees, on its own responsibility, to solve and settle the claims and damages and to defend, indemnify or hold Yokogawa totally harmless, from or against any liabilities, losses, damages and expenses (including fees for recalling the Products and reasonable attorney's fees and court costs), or claims arising out of and related to the above-said claims and damages.

Article 4 (Warranty)

- 4.1The Licensee shall agree that the Licensed Software shall be provided to the Licensee on an "as is" basis when delivered. If defect(s), such as damage to the medium of the Licensed Software, attributable to Yokogawa is found, Yokogawa agrees to replace, free of charge, any Licensed Software on condition that the defective Licensed Software shall be returned to Yokogawa's specified authorized service facility within seven (7) days after opening the Package at the Licensee's expense. As the Licensed Software is provided to the Licensee on an "as is" basis when delivered, in no event shall Yokogawa warrant that any information on or in the Licensed Software, including without limitation, data on computer programs and program listings, be completely accurate, correct, reliable, or the most updated.
- 4.2Notwithstanding the preceding Paragraph 4.1, when third party software is included in the Licensed Software, the warranty period and terms and conditions that apply shall be those established by the provider of the third party software.
- 4.3When Yokogawa decides in its own judgement that it is necessary. Yokogawa may from time to time provide the Licensee with Revision upgrades and Version upgrades separately specified by Yokogawa (hereinafter called "Updates").
 4.4Notwithstanding the preceding Paragraph 4.3, in no event shall Yokogawa provide Updates where the Licensee or any third party conducted renovation or improvement of the Licensee Software.
 4.THE CONFERENTIAL OF EXCLUSIVE AND IN USE AND IN USE AND CONFERENTIAL OF AND INTER AND PROPERTY AND PROPERTY AND PROPERTY.
- 4.5THE FOREGOING WARRANTIES ARE EXCLUSIVE AND IN LIEU OF ALL OTHER WARRANTIES OF QUALITY AND PERFORMANCE, WRITTEN, ORAL, OR 4.5 THE FOR EQUING WARRANTIES INCLUDING ANY IMPLIED WARRANTIES OF MOALTH FINES FOR A PARTICULAR PURPOSE ARE IMPLIED, AND ALL OTHER WARRANTIES INCLUDING ANY IMPLIED WARRANTIES OF MERCHANTABILITY OF FINESS FOR A PARTICULAR PURPOSE ARE HEREBY DISCLAIMED BY YOKOGAWA AND ALL THIRD PARTIES LICENSING THIRD PARTY SOFTWARE TO YOKOGAWA.
 4.6 Correction of nonconformity in the manner and for the period of time provided above shall be the License's sole and exclusive remedy for any failure of Yokogawa to
- comply with its obligations and shall constitute fulfillment of all liabilities of Yokogawa and any third party licensing the Third Party Software to Yokogawa (including any liability for direct, indirect, special, incidental or consequential damages) whether in warranty, contract, tort (including negligence but excluding willful conduct or gross negligence by Yokogawa) or otherwise with respect to or arising out of the use of the Licensed Software.

Article 5 (Infringement)

- 5.1lf and when any third party should demand injunction, initiate a law suit, or demand compensation for damages against the Licensee under patent right (including utility model right, design patent, and trade mark), copy right, and any other rights relating to any of the Licensed Software, the Licensee shall notify Yokogawa in writing to that effect without delay.
- Such the case of the preceding Paragraph 5.1, the Licensee shall assign to Yokogawa all of the rights to defend the Licensee and to negotiate with the claiming party. Furthermore, the Licensee shall provide Yokogawa with necessary information or any other assistance for Yokogawa's defense and negotiation. If and when such a claim should be attributable to Yokogawa, subject to the written notice to Yokogawa stated in the preceding Paragraph 5.1, Yokogawa shall defend the Licensee and negotiate with the claiming party at Yokogawa's cost and expense and be responsible for the final settlement or judgment granted to the claiming party in the preceding Paragraph 5.1.
- 5.3When any assertion or allegation of the infringement of the third party's rights defined in Paragraph 5.1 is made, or when at Yokogawa's judgment there is possibility of such assertion or allegation, Yokogawa will, at its own discretion, take any of the following countermeasures at Yokogawa's cost and expense. a) To acquire the necessary right from a third party which has lawful ownership of the right so that the Licensee will be able to continue to use the Licensed Software; b) To replace the Licensed Software with an alternative one which avoids the infringement; or

 - To remodel the Licensed Software so that the Licensed Software can avoid the infringement of such third party's right.
- 5.4lf and when Yokogawa fails to take either of the countermeasures as set forth in the preceding subparagraphs of Paragraph 5.3, Yokogawa shall indemnify the Licensee only by paying back the price amount of the Licensed Software which Yokogawa has received from the Licensee. THE FOREGOING PARAGRAPHS STATE THE ENTIRE LIABILITY OF YOKOGAWA AND ANY THIRD PARTY LICENSING THIRD PARTY SOFTWARE TO YOKOGAWA WITH RESPECT TO INFRINGEMENT OF THE INTELLECTUAL PROPERTY RIGHTS INCLUDING BUT NOT LIMITED TO, PATENT AND COPYRIGHT.
- 5.5Regardless of this article, the terms and conditions provided separately apply to demands for compensation or the like arising from the use of the Third-Party Software or OSS.

Terms and Conditions of the Software License

Article 6 (Liabilities)

- 6.1ff and when the Licensee should incur any damage relating to or arising out of the Licensed Software or service that Yokogawa has provided to the Licensee under the conditions herein due to a reason attributable to Yokogawa, Yokogawa shall take actions in accordance with this Agreement. However, in no event shall Yokogawa be liable or responsible for any special, incidental, consequential and/or indirect damage, whether in contract, warranty, tort, negligence, strict liability, or otherwise, including, without limitation, loss of operational profit or revenue, loss of use of the Licensed Software, or any associated products or equipment, cost of capital, loss or cost of interruption of the Licensee's business, substitute equipment, facilities or services, downtime costs, delays, and loss of business information, or claims of customers of Licensee or other third parties for such or other damages. Even if Yokogawa is liable or responsible for the damages attributable to Yokogawa has received. Please note that Yokogawa shall be released or discharged from part or all of the liability under this Agreement if the Licensee modifies, remodels, combines with other software or products, or causes any deviation from the basic specifications or functional specifications, without Yokogawa's prior written consent.
 6.2All causes of action against Yokogawa arising out of or relating to this Agreement or the performance or breach hereof shall expire unless Yokogawa is notified of the claim within one (1) year of its occurrence.
- 6.3In no event, regardless of cause, shall Yokogawa assume responsibility for or be liable for penalties or penalty clauses in any contracts between the Licensee and its customers.

Article 7 (Limit of Export)

Unless otherwise agreed by Yokogawa, the Licensee shall not directly or indirectly export or transfer the Licensed Software to any countries other than those where Yokogawa permits export in advance.

Article 8 (Term)

This Agreement shall become effective on the date when the Licensee receives the Licensed Software and continues in effect unless or until terminated as provided herein, or the Licensee ceases using the Licensed Software by itself or with Yokogawa's thirty (30) days prior written notice to the Licensee.

Article 9 (Injunction for Use)

During the term of this Agreement, Yokogawa may, at its own discretion, demand injunction against the Licensee in case that Yokogawa deems that the Licensed Software is used improperly or under severer environments other than those where Yokogawa has first approved, or any other condition which Yokogawa may not permit.

Article 10 (Termination)

Yokogawa, at its sole discretion, may terminate this Agreement without any notice or reminder to the Licensee if the Licensee violates or fails to perform this Agreement. However, Articles 5, 6, and 11 shall survive even after the termination.

Article 11 (Jurisdiction)

Any dispute, controversies, or differences between the parties hereto as to interpretation or execution of this Agreement shall be resolved amicably through negotiation between the parties upon the basis of mutual trust. Should the parties fail to agree within ninety (90) days after notice from one of the parties to the other, both parties hereby irrevocably submit to the exclusive jurisdiction of the Tokyo District Court (main office) in Japan for settlement of the dispute.

Article 12 (Governing Law)

This Agreement shall be governed by and construed in accordance with the laws of Japan. The Licensee expressly agrees to waive absolutely and irrevocably and to the fullest extent permissible under applicable law any rights against the laws of Japan which it may have pursuant to the Licensee's local law.

Article 13 (Severability)

In the event that any provision hereof is declared or found to be illegal by any court or tribunal of competent jurisdiction, such provision shall be null and void with respect to the jurisdiction of that court or tribunal and all the remaining provisions hereof shall remain in full force and effect.

Article 14 (Change in Agreement)

The contents of this contract, contents of this product, specifications of the licensed software, related support and other related services can be changed at Yokogawa's discretion.

Checking the Contents of the Package

Unpack the box and check that the contents below are available. If some of the contents are not correct or missing or if there is physical damage, contact the dealer from which you purchased them.

• AQ6370 Viewer Installation Manual (IM 735371-02E)

CD-ROM

The CD-ROM contains the following items.

• AQ6360 Viewer (software for AQ6360, included with CD-ROM Edition 14 or later)

- AQ6370 Viewer (software for AQ6370)
- AQ6370B Viewer (software for AQ6370B, included with CD-ROM Edition 5 or later)
- AQ6370C Viewer (software for AQ6370C, included with CD-ROM Edition 7 or later)
- AQ6370D Viewer (software for AQ6370D, included with CD-ROM Edition 9 or later)
- AQ6373 Viewer (software for AQ6373, included with CD-ROM Edition 6 or later)
- AQ6373B Viewer (software for AQ6373B, included with CD-ROM Edition 10 or later)
- AQ6374 Viewer (software for AQ6374, included with CD-ROM Edition 13 or later)
- AQ6375/AQ6375B/AQ6376 Viewer (software for AQ6375, included with CD-ROM Edition 12 or later)
- AQ6377 Viewer (software for AQ6377, included with CD-ROM Edition 18 or later)
- OSA Viewer (common software, included with CD-ROM Edition 14 or later)
- AQ6370 Firmware Updater
- The updater is used when the AQ6370 firmware version is older than R02.01.
- AQ6370 Viewer User's Manual (this manual)

• License sheet (1)

You must enter the license code indicated on the license sheet when you start the software program. For details, see section 2.1, "Starting and Closing the Software Program."

System Requirements and Notes on Using the Software Program

PC System Requirements

PC

A PC running following operating system with at least 256 MB of memory (512 MB or more recommended).

Operating System

- Windows Vista (32 bit/64 bit) (Service Pack 2 or later)
- Windows 7 (32 bit/64 bit) (Service Pack 1 or later)
- Windows 8/8.1 (32 bit/64 bit)
- Windows 10 (32 bit/64 bit)

CD-ROM Drive

A CD-ROM drive is required to install this software program.

Ethernet interface

A LAN port (100BASE-TX) is required to use the Optical Spectrum Analyzer in remote control mode or file transfer mode.

Display, Mouse, and Printer

- Display: 1024 × 768 pixels, 65536 colors or more
- Mouse and printer: Those compatible with the operating systems above (the printer is required when printing a hard copy of the screen)

Notes on Using the Software Program

Storing the CD-ROM

Please store the original CD-ROM (containing the software program) in a safe place. During the actual operation, use the software program that is installed on the hard disk.

Precautions to Be Taken When Using the Software Program

- Do not operate the Optical Spectrum Analyzer from the panel while you are controlling the Optical Spectrum Analyzer using this software program. If you do, operation errors can result.
- Only a single Optical Spectrum Analyzer can be controlled by this software program. In addition, simultaneous connection to the Optical Spectrum Analyzer from multiple PCs is not allowed.
- Limitations exist such as in the result of the analysis function if a waveform file other than that of the AQ6360/AQ6370/AQ6370B/AQ6370C/AQ6370D/AQ6373/ AQ6373B/AQ6374/AQ6375/AQ6375B/AQ6376/AQ6377 is loaded into this software program. For details, see page 1-5.

Installing and Uninstalling the Software Program

Procedure

Installing the Software Program

The CD-ROM contains viewers corresponding to the various optical spectrum analyzer models.

Select the viewer you want to install from the installation tool that starts automatically. If the installation tool does not start automatically, run Autorun.exe in the CD.

- To install AQ6360 Viewer: AQ6360 Viewer
- To install AQ6370 Viewer: AQ6370 Viewer
- To install AQ6370B Viewer: AQ6370B Viewer
- To install AQ6370C Viewer: AQ6370C Viewer
- To install AQ6370D Viewer: AQ6370D Viewer
- To install AQ6373 Viewer: AQ6373 Viewer
- To install AQ6373B Viewer: AQ6373B Viewer
- · To install AQ6374 Viewer: AQ6374 Viewer
- To install AQ6375/AQ6375B/AQ6376 Viewer: AQ6375&76 Viewer
- To install AQ6377 Viewer: AQ6377 Viewer
- To install OSA Viewer: OSA Viewer

The procedure below is for installing the software program on Windows 7. AQ6370 Viewer is used for this example. For other viewer, substitute "AQ6360/AQ6370B/ AQ6370C/AQ6370D/AQ6373/AQ6373B/AQ6375/AQ6375B/AQ6376/OSA/AQ6377" wherever "AQ6370" appears.

- Start Windows and log on with administrator privileges. If AQ6370 Viewer is already installed, uninstall it first.
- **2.** Load the disc into the CD-ROM drive. The installation tool starts automatically.
- 3. Select AQ6370 Viewer, and then click Select. The installation begins



4. Press the Next button.



5. Read the license agreement carefully. If you agree to the terms, click the **I Agree** option button and then click **Next**.



6. Select the installation destination drive and folder, and click Next. The installation destination setup screen shown below opens. The default installation destination is C:\Program Files\Yokogawa\.

JJ AQ6370Viewer	
Select Installation Folder	Yokogawa 🔶
The installer will install AQ6370Viewer to the following folder. To install in this folder, click "Next". To install to a different fo	
Eolder: C:\Program Files\Yokogawa\AQ6370Viewer\	B <u>r</u> owse
	Disk Cost
Install AQ6370Viewer for yourself, or for anyone who	uses this computer:
─ Everyone	
⊚ Just <u>m</u> e	
Cancel	<back next=""></back>

7. In the installation destination confirmation screen, click Next. The software installation starts. A progress bar is shown in the screen while the installation is in progress.



8. Click Close to finish the installation.

Yokogawa > AQ6370 Viewer is added to Programs in the Start menu in the task bar, and a AQ6370 Viewer shortcut icon is created on the desktop.

Jacobs AQ6370Viewer	
Installation Complete	Yokogawa ┥
AQ6370Viewer has been successfully installed.	
Cancel	< Back

Note.

- When you run the AQ6370 Viewer for the first time, a license confirmation screen opens. Click Input License code, and enter the license code indicated in the license seal to open the main screen. For details, see section 2.1, "Starting and Closing the Software Program."
- AQ6360 Viewer is included with CD-ROM Edition 14 or later.
- AQ6370B Viewer is included with CD-ROM Edition 5 or later.
- AQ6370C Viewer is included with CD-ROM Edition 7 or later.
- AQ6370D Viewer is included with CD-ROM Edition 9 or later.
- AQ6373 Viewer is included with CD-ROM Edition 6 or later.
- AQ6373B Viewer is included with CD-ROM Edition 10 or later.
- AQ6374 Viewer is included with CD-ROM Edition 13 or later.
- AQ6375/AQ6375B/AQ6376 Viewer is included with CD-ROM Edition 12 or later.
- · AQ6377 Viewer is included with CD-ROM Edition 18 or later.
- OSA Viewer is included with CD-ROM Edition 14 or later.
- Multiple different models of Viewers can be installed on the same PC.

Uninstalling the Software Program

The procedure below is for uninstalling the software program on Windows 7.

- 1. On the task bar, click the Start button and choose Control Panel.
- 2. Double-click Uninstall program of Programs from the Control Panel.
- *3.* In the Programs and Features window, select AQ6370 Viewer, and click **Uninstall**. A confirmation dialog box for deleting the program opens.
- 4. Click Yes.

The software program is uninstalled.

Note _

The files that the user saved (waveform files, etc.) remain in the follwing folder even if you uninstall the software program.

My Document\AQ6370Viewer Data

Contents

	Terms and Conditions of the Software License					
	Checking the Contents of the Package	iv				
	System Requirements and Notes on Using the Software Program	v				
	Installing and Uninstalling the Software Program	vi				
Chapter 1	Functional Overview					
•	1.1 Operation Modes	1-1				
	1.2 Functional Difference in Operation Modes and Loadable Files					
	1.3 Functions of the Parts of the Screen					
Chapter 2	Basic Operations					
-	2.1 Starting and Closing the Software Program	2-1				
	2.2 Basic Operation Using the Mouse and Keys	2-3				
	2.3 Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer	2-6				
Chapter 3	Common Operations in Viewer Mode and Remote Control Mode (for					
	the model-specific software)					
	3.1 Main Menu					
	3.2 Making a Hard Copy of the Screen	3-4				
	3.3 Setting the User Folder	3-5				
Chapter 4	Operations in Viewer Mode (for the model-specific software)					
	4.1 Loading Waveform Data and Displaying Waveforms	4-1				
	4.2 Saving the Setup Data	4-4				
	4.3 Soft Key Menu in Viewer Mode	4-8				
	4.4 Commands That Do Not Operate in the Program Function	4-9 In				
Chapter 5	Operations in Remote Control Mode (for the model-specific					
	software)					
	5.1 Connect and Disconnect Operations	5-1				
	5.2 Precautions to Be Taken When Operating in Remote Control Mode					
	5.3 Soft Key Menu in Remote Control Mode					
Chapter 6	File Transfer					
•	6.1 File Transfer Operation	6-1				
Chapter 7	Operation of the Common Software					
	7.1 Start the Common Software	7-1				
	7.2 Operations on the Viewer Software (Common software)					
	7.3 Operations on the Remote Control Software (Common software)					
Chapter 8	Troubleshooting					
	8.1 Troubleshooting	8-1				
	8.2 Messages and Corrective Actions					
Index						
IIIUEX						

1

ex

1

1.1 Operation Modes

Model-specific Software and Common Software

This software product contains two types of software:

- · Model-specific software
- Common software (OSA Viewer)

Model-Specific Software

The model-specific software has the following three operation modes.

The relationship between these modes will be explained using the AQ6370 Viewer window as an example.

An AQ6370 Viewer screen is used for this example.

The software application has the three operation modes listed below.

The program starts in Viewer mode.

The mode switches to Remote Control if you connect to the Optical Spectrum Analyzer. Select FileTransfer in the main menu to switch to File Transfer mode.



Viewer Mode

The Optical Spectrum Analyzer display is shown on the PC display. You can use the mouse and keyboard to operate the display in the same manner as with the Optical Spectrum Analyzer.

You can carry out the following operations in Viewer mode.

 Load the waveform data that was saved on the AQ6360/AQ6370/AQ6370B/AQ6370C/ AQ6370D/AQ6373/AQ6373B/AQ6374/AQ6375/AQ6375B/AQ6376/AQ6377 to the USB storage device and display the waveform in the screen.

Waveform data saved on the AQ6317 or AQ6319 can also be loaded.

- Perform marker operations and analysis on the displayed waveform and display the result.
- · Create and edit programs and template data.
- Save the created data to a file.



1.1 Operation Modes

Display Examples



Remote Control Mode

The same display as the display on the Optical Spectrum Analyzer connected via the Ethernet interface is shown on the PC display. You can use the mouse and keyboard to operate the display in the same manner as with the Optical Spectrum Analyzer unit. You can carry out the following operations in Remote Control mode.

- Set the measurement conditions on the target Optical Spectrum Analyzer and execute measurements.
- Display the waveform data that is measured on the Optical Spectrum Analyzer on the PC display.
- Perform marker operations and analysis on the displayed waveform and display the result.



Display Examples



Note

The waveform data that can be loaded in Remote Control mode is the data stored on the PC in which this software program is installed. In addition, waveform data can be saved in Remote Control mode only to the PC in which this software program is installed.

File Transfer Mode

Files that are stored in the Optical Spectrum Analyzer can be transferred at high speeds to the PC via the Ethernet interface. Files also be transferred from the PC to the Optical Spectrum Analyzer.



Common Software (OSA Viewer)

When you start OSA Viewer, the top menu appears. For this top menu, you can start various software applications such as Viewer, Remote Control, and File Transfer. These applications provide functions that are nearly equivalent as the Viewer, Remote Control, and File Transfer modes of the model-specific software.



OSA Viewer supports the AQ6360 Series (AQ6360), AQ6370 Series (AQ6370D/ AQ6370C), AQ6373 Series (AQ6373/AQ6373B), AQ6375 Series (AQ6375/AQ6375B), AQ6373 Series (AQ6374), AQ6376 Series (AQ6376) and AQ6377 Series (AQ6377). It does not support the AQ6370 or AQ6370B.

Functional Differences in Operation Modes

The functions that you can execute vary depending on the operation mode as follows:

Viewer Mode

Almost all the functions except the sweep function can be executed. Some commands do not operate in the execution of the program function.

Remote Control Mode

Almost all the functions except the program function can be executed. However, the waveform displayed on the Optical Spectrum Analyzer and the waveform displayed on the PC display may not match during remote control.

• File Transfer Mode Only file transfer between the Optical Spectrum Analyzer and the PC is possible.

Main Operation Differences in Viewer and Remote Control Modes

For details, see section 4.3, "Soft Key Menu in Viewer Mode" and section 5.3, "Soft Key Menu in Remote Control Mode."

Front Panel Key	Viewer	Remote Control	I Note	
Menu	Mode	Mode		
CENTER	Yes	Yes		
SPAN	Yes	Yes		
LEVEL	Yes	Yes		
SWEEP	No	Yes	AUTO, REPEAT, SINGLE, STOP, and SEGMENT MEASURE cannot be used in Viewer mode.	
ZOOM	Yes	Limit	Waveform magnification and reduction can be carried out only on the PC in Remote Control mode.	
SETUP	Yes	Yes		
TRACE	Yes	Limit	Waveform does not match between the Optical Spectrum Analyzer and PC when using ROLL AVG, MAX HOLD, or MIN HOLD in Remote Control mode.	
DISPLAY	Limit	Limit	DISPLAY OFF cannot be used in Viewer mode. Waveform does not match between the Optical Spectrum Analyzer and PC when using NOISE MASK operation in Remote Control mode.	
MARKER	Yes	Limit	Marker function can only be used on the PC in Remote Control mode (line markers 1 and 2 can be used on the Optical Spectrum Analyzer).	
PEAK SEARCH	Yes	Limit	The search mode setting and multi search function can only be executed on the PC in Remote Control mode.	
ANALYSIS	Yes	Limit	The analysis function can only be executed on the PC in Remote Control mode.	
USER	Yes	Yes		
MEMORY	Yes	Yes		
FILE	Yes	Yes		
PROGRAM	Limit	Limit	Editing is allowed only on the PC. Some commands cannot be executed in Viewer mode. Programs cannot be executed in Remote Control mode.	
ADVANCE	Limit	Limit	DATA LOGGING > START cannot be used in Viewer mode. DATA LOGGING cannot be used in Remote Control mode. Template editing and execution are allowed on the PC in Remote Control mode.	
SYSTEM	Limit	Limit	OPTICAL ALIGNMENT, WL CALIBRATION, SYSTEM INFORMATION, and RES BW CALIBRATION cannot be performed in Viewer mode. The progress of OPTICAL ALIGNMENT, WL CALIBRATION, and RES BW CALIBRATION is not displayed in Remote Control mode.	
UNDO/LOCAL	No	Yes		
COPY	Yes	Yes		
FEED	No	No	AQ6370, AQ6370B, AQ6370C, AQ6375, AQ6373	
PRESET	Yes	Yes	Excludes above	
HELP	Yes	Yes		

Yes: Available, No: Not available, Limit: With limitation.

Loadable Files

In Viewer or Remote Control mode, the files listed below that have been stored on the AQ6360, AQ6370, AQ6370B, AQ6370C, AQ6370D, AQ6373, AQ6373B, AQ6375, AQ6375B, AQ6376, AQ6377, AQ6317, or AQ6319 can be loaded. The extensions of the loadable data are given in parentheses.

AQ6360 Viewer

- Waveform data (.csv, wva, wx9, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wva, wx9, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.sta, .st9, .st8, .st7, .st6, .st5, and .st4)
- Program data (.pga, .pg9, .pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dta, .dt9, .dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

AQ6370 Viewer

- Waveform data (.csv, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st6, .st5, and .st4)
- Program data (.pg6, .pg5, and .pg4)
- Analysis result data (.dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

AQ6370B Viewer

- Waveform data (.csv, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st8, .st7, .st6, .st5, and .st4)
- Program data (.pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

AQ6370C Viewer and AQ6370D Viewer

- Waveform data (.csv, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st8, .st7, .st6, .st5, and .st4)
- Program data (.pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)
- Logging data (.lg8)

AQ6373 Viewer

- Waveform data (.csv, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st9, .st8, .st7, .st6, .st5, and .st4)
- Program data (.pg9, .pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt9, .dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

AQ6373B Viewer

- Waveform data (.csv, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st9, .st8, .st7, .st6, .st5, and .st4)
- Program data (.pg9, .pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt9, .dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)
- Logging data (.lg9, .lg8)

AQ6374 Viewer

- Waveform data (.csv, .wx9, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wx9, .wv9, .wv8, wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.sx9, .st9, .st8, .st7, .st6, .st5, and .st4)
- Program data (.pg9, .pg8, .pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt9, .dt8, .dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)
- Logging data (.lx9, .lg9, .lg8)

AQ6375/AQ6375B/AQ6376 Viewer

- Waveform data (.csv, .wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st7, .st6, .st5, and .st4)
- Program data (.pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

AQ6377 Viewer

- Waveform data (.csv, .wv7, .wv6, .wv5, .wv4, and .txt)
- Memory data (.csv, .wv7, .wv6, .wv5, .wv4, and .txt)
- Setup data (.st7, .st6, .st5, and .st4)
- Program data (.pg7, .pg6, .pg5, and .pg4)
- Analysis result data (.dt7, .dt6, .dt5, .dt4, and .csv)
- Template data (.csv)

Note .

- If AQ6317 or AQ6319 data are loaded, some functions can not be executed.
- To analyze waveform data from the AQ6360, AQ6370, AQ6370B, AQ6370C, AQ6370D, AQ6373, AQ6373B, AQ6374, AQ6375, AQ6375B, AQ6376 or AQ6377 use the model's corresponding Viewer.

Otherwise, the following analysis results and the displayed contents cannot be guaranteed. (including AQ6317 or AQ6319 waveform data)

Power analysis, LED analysis, WDM analysis, OSNR (WDM) analysis, EDFA-NF analysis, and switching of the vertical axis unit to dBm/nm

Connectable AQ6370s and Firmware Update

Main unit firmware updates are sometimes required with the AQ6370. With the AQ6360/AQ6370B/AQ6370C/AQ6370D/AQ6373/AQ6373B/AQ6374/AQ6375/ AQ6375B/AQ6376/AQ6377, updates are not required. Connection can be made regardless of the version.

AQ6370s required updates

If the firmware version is older than R02.01, update the firmware using the AQ6370 Firmware Updater in the UPDATE folder (\UPDATEFILE\AQ6370) on the accompanying CD-ROM.

The AQ6370s that can be connected in Remote Control mode and File Transfer mode are those with firmware version R02.01 or later.

Checking the Firmware Version

- 1. Press SYSTEM on the AQ6370.
- 2. Press the MORE soft key to display MORE 4/4.
- 3. Press the VERSION soft key.

Update Procedure

- 1. Insert the accompanying CD-ROM and a USB memory into the PC.
- 2. Copy the UPDATE folder on the CD-ROM to the USB memory.
- **3.** Insert the USB memory into the AQ6370.
- Update the firmware according to the procedure given in section 10.1, "Upgrading the Firmware" in the AQ6370 Optical Spectrum Analyzer User's Manual (IM 735301-01E).

Note

If the firmware version of the AQ6370 is already R02.01 or later, do not update the firmware.

1.3 Functions of the Parts of the Screen

The figure below shows the screen in Viewer mode. The same screen as that of the actual instrument is displayed. An AQ6370 Viewer screen is used for this example.



No. Function

Data area

1

- 2 Measurement condition area
- 3 Displays NEW when the measurement conditions are changed.
- 4 Displays the level axis scale per division
- 5 Displays \bigcup_{CAL}^{UN} when the settings for span, the number of sampling points, and the resolution are inappropriate.
- 6 Displays the reference level
- 7 Label area (up to 56 characters)
- 8 Displays the resolution
- 9 Displays the measurement sensitivity
- 10 Displays the average count
- 11 Displays the number of samples
- 12 Displays the year, month, date, and hour
- 13 Displays the status of each trace.
- 14 Displays ZOOMING when using the ZOOM function

Displays the typical setup status				
If a setting is turned ON, the button is shown depressed. If the screen is black				
and white, it is shown in black.				
TLS SYCON/OFF setting of TLS SNC SWEEP (Not displayed when using AQ6373 Viewer)				
COR SIZSMALL/LARGE setting of FIBER CORE SIZE (Only displayed when using				
AQ6373 Viewer)				
RES CORON/OFF setting of RESOLN CORRECT				
LVL SHFON/OFF setting of LEVEL SHIFT				
WF SHFON/OFF setting of WF SHIFT				
NOI MSKNOISE MASK setting SRC ZOMON/OFF setting of SEARCH/ANA ZOOM AREA				
SRC 1-2ON/OFF setting of SEARCH/ANA L1-L2				
VAC WLMEAS WL setting				
AUT OFSON/OFF setting of AUTO OFFSET				
AUT ANAON/OFF setting of AUTO ANALYSIS				
AUT SRCON/OFF setting of AUTO SEARCH				
AUT SCLON/OFF setting of AUTO SUB SCALE AUT REFON/OFF setting of AUTO REF LEVEL				
AUT CTRON/OFF setting of AUTO CENTER				
SWP INTON/OFF setting of SWEEP INTERVAL (Not displayed when using AQ6373				
Viewer)				
SWP 1-2ON/OFF setting of SWEEP MKR L1-L2				
SMO OTHON/OFF setting of SMOOTHING (Only displayed when using AQ6373				
Viewer) RPTREPEAT (repeat sweep)				
SGLSINGLE (single sweep)				
SPTSTOP (sweep stop)				
Displays the wave length axis scale per division				
Displays the sweep status				
RPT = Repeat, SGL = Single, and STP = Stop				
Displays the soft key menu				
Parameter value display area				
Parameter value display area Parameter value input area OVERVIEW display screen				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function.				
Parameter value display area Parameter value input area OVERVIEW display screen				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software.				
 Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program 				
 Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. 				
 Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to disconnect the optical spectrum analyzer. 				
 Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. 				
 Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." File				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to disconnect the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. File TransferSelect to set the folder for saving the loaded data. DisplaySelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only)				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the commonsoftware. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect the optical spectrum analyzer. DisconnectSelect to transfer files with the optical spectrum analyzer. File TransferSelect to set the folder for saving the loaded data. DisplaySelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect the optical spectrum analyzer. DisconnectSelect to transfer files with the optical spectrum analyzer. File TransferSelect to disconnect the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. DisconnectSelect to disconnect the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software For details on the main menu, see section 7.1, "Main Menu."				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actuat waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software For details on the main menu, see section 7.1, "Main Menu." Open File:Select this to select the waveform data to be displayed.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to disconnect the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software For details on the main menu, see section 7.1, "Main Menu." Open File:Select this to select the waveform data to be displayed. Save Screen: .Select this to save the displayed screen as image data.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. DisconnectSelect to disconnect the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actua waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software For details on the main menu, see section 7.1, "Main Menu." Open File:Select this to select the waveform data to be displayed. Save Screen: .Select this to print a hard copy of the screen.				
Parameter value display area Parameter value input area OVERVIEW display screen Displayed only when using the ZOOM function. Displays the sub scale Main menu The main menu is different between the model-specific software and the common software. Model-specific software For a detailed description of the main menu, see section 3.1, "Main Menu." FileSelect to output a hard copy of the screen or when closing the software program ConnectSelect to connect to the optical spectrum analyzer. DisconnectSelect to disconnect the optical spectrum analyzer. File TransferSelect to transfer files with the optical spectrum analyzer. SettingSelect to display the model that measured the waveform data and the actual waveform resolution (AQ6375/AQ6375B/AQ6376 Viewer only) AboutSelect to check the software version. Common software For details on the main menu, see section 7.1, "Main Menu." Open File:Select this to select the waveform data to be displayed. Save Screen: .Select this to save the displayed screen as image data.				

1.3 Functions of the Parts of the Screen

- 24 Front panel key menu
 - A menu that is shown when you right-click the mouse. The menu is equivalent to the front panel keys of the optical spectrum analyzer.
- 25



(Turns green when a command or data is transmitted or received)

2.1 Starting and Closing the Software Program

Procedure

An AQ6370 Viewer screen is used for this example.

Starting the Software of Each Model

On the task bar, click the Start button, point to **Programs > Yokogawa > AQ6370 Viewer** (AQ6360 Viewer/AQ6370B Viewer/AQ6370C Viewer/AQ6370D Viewer/AQ6373 Viewer/ AQ6373B Viewer/AQ6374 Viewer/AQ6375&76 Viewer/AQ6377) or double-click the AQ6370 Viewer/AQ6360 Viewer/AQ6370B Viewer/AQ6370C Viewer/AQ6370D Viewer/ AQ6373 Viewer/AQ6373B Viewer/AQ6374 Viewer/AQ6375&76/AQ6377 Viewer shortcut icon.

When the program starts, the main windows shown below opens. The window shows the contents of the window that was shown when the program was closed the last time.



License Authentication

When the program is started for the first time, a license authentication dialog box opens as shown below.

📕 License Message			
This product is un-licensed.			
There are 30 days remaining in your evaluation period.			
<u>()</u>	Input License Code		

In the License Message dialog box, click **Input License Code**. In the Input License Code dialog box that opens, type the user name, organization (optional), and license code. The license code is indicated on the accompanying CD case or the license sheet.

🗰 Input License Code	×
User Name:	
Organization:	
License Code:	
Cancel	

2.1 Starting and Closing the Software Program

When the license is authenticated, the message below appears. Click **OK** to show the main window.

Thank you	×
The license code has been accepted.	
Thank you for using the AQ6370Viewer.	
CK I	

Note .

If the license code is not correct, an error message appears. If this happens, click **OK** and retype the license code again in the Input License Code dialog box.

Exiting the Software

On the File menu, choose Exit.

Starting the Common Software (OSA Viewer)

On the Windows start menu, click Programs > Yokogawa > OSA Viewer, or double-click the OSA Viewer icon on the desktop.

A navigation window appears.



When you start the viewer software, a license authentication window appears, as with the model-specific software. Proceed in the same manner as with the model-specific software.

For instructions on how to use the common software, see chapter 7.

Exiting the Software

Click X in the upper right of the window.

2.2 Basic Operation Using the Mouse and Keys

Mouse Operation

Displaying the Front Panel Key Menu

Right-click in the window. The front panel key names in the FUNCTION section of the Optical Spectrum Analyzer appear.

Optical Spectrum
CENTER
SPAN
LEVEL
SWEEP
ZOOM
SETUP
TRACE
DISPLAY
MARKER
PEAK SEARCH
ANALYSIS
USER
MEMORY
FILE
PROGRAM
ADVANCE
SYSTEM
UNDO/LOCAL
COPY
FEED*
HELP

* PRESET on the AQ6360 Viewer, AQ6370D Viewer, the AQ6373B Viewer, AQ6374 Viewer, the AQ6375&76 Viewer and the AQ6377 Viewer

Choosing a Menu Command and Closing the Front Panel Key Menu

Left-click the desired menu command to switch to the corresponding soft key menu. If you choose a menu command, the front panel key menu clears. Clicking outside the front panel key menu also closes the menu.

Choosing a Soft Key and Operating the Parameter Value Display/Input Area

Left-click the desired soft key menu command.

If you choose a soft key that requires you to enter a value, a parameter display/input area appears.You can change the parameter value by clicking the keys in the input area as described below.

Increase or decrease the input value



Note -

- If the value entered using the numeric keys do not match with allowable values, it is set to the closest allowable value.
- If you click the BS key repeatedly and the input value display area becomes empty, the input value area clears, and the screen returns to the condition before you started entering the values.
- If you want to set OFF on a parameter that can be set to OFF, hold down the $\bigtriangledown\,\mathsf{key}.$

Keyboard Operation

You can also use the PC keyboard to carry out operations as with the mouse. For the mapping of the panel keys to the keyboard, see the Panel Key Mapping Table below.

In addition, you can directly enter labels, file names, values, and the like from the keyboard.

Group		Function Name	Keyboard	Description
Sweep		SWEEP	[SHIFT]+[F1]	Execute/Start the sweep.
	Measurement condition settings	CENTER	[SHIFT]+[F2]	Set the measurement center wavelength.
		SPAN	[SHIFT]+[F3]	Set the measurement span.
		LEVEL	[SHIFT]+[F4]	Set the level axis.
		SETUP	[SHIFT]+[F5]	Set the resolution, sensitivity, etc.
	Display	TRACE	[SHIFT]+[F6]	Set the trace.
	condition	ZOOM	[SHIFT]+[F7]	Set the display scale.
Setting or	settings	DISPLAY	[SHIFT]+[F8]	Set the screen display.
execution of each		MARKER	[SHIFT]+[F9]	Set the marker.
function	Analysis	SEARCH	[SHIFT]+[F10]	Peak/bottom search function
		ANALYSIS	[SHIFT]+[F11]	Set the analysis function.
		USER	[ALT]+[F1]	User-defined menu
		MEMORY	[ALT]+[F2]	Memory
	Miscellaneous	FILE	[ALT]+[F3]	Save to file, load from file, and other file operations.
		PROGRAM	[ALT]+[F4]	Program function.
		ADVANCE	[ALT]+[F5]	Expansion function
		SYSTEM	[ALT]+[F6]	Set the system.
Soft keys		F1 to F9	F1 to F9	Varies depending on the menu.
		UNDO/LOCAL	[ALT]+[F9]	Undo function.
Auxiliary ke	evs	COPY	[ALT]+[F10]	Screen copy.
Auxiliary keys		HELP	[ALT]+[F12]	Display help (UNDO/LOCAL key to close help)
		Numeric keys	0123456789	Enter values.
		BACK SPACE	Back Space	Backspace.
		µm/ENTER	None	Enter the input.
Parameter value input		nm/ENTER	ENTER	Enter the input.
		Rotary knob	$[\rightarrow]$ and $[\leftarrow]$	Change the value or parameter.
		Arrow keys (▲ ▼)	[↑] and [↓]	Change the value step-wise, change the parameter, and scroll the table.
		COARSE	[ALT]+[N]	Switch between fine and coarse setting of the encoder.

2.2 Basic Operation Using the Mouse and Keys

Soft Key Menu Shapes and Functions

The soft key menu comes in different shapes to indicate its functionality.



Note_

The shapes shown on the last menu command (see below) are not soft keys. They indicate a front panel key name or the soft key menu hierarchy.



Y SCALE SETTING

Indicates the soft key menu hierarchy. This indicates that the soft key menu is under the Y SCALE SETTING hierarchy.

2.3 Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer

Connection Procedure

Connect a UTP (Unshielded Twisted-Pair) cable or an STP (Shielded Twisted-Pair) cable that is connected to a hub, for example, to the 100BASE-TX port on the rear panel of the Optical Spectrum Analyzer.



Precautions to Be Taken When Connecting the Cable

- Be sure to use a straight cable via a hub for the connection between the Optical Spectrum Analyzer and the PC.
- If you are using a UTP cable (straight cable), use a cable of category 5.

Setting the Ethernet Interface

If you are using Remote Control mode or File Transfer mode, set the Ethernet interface of the Optical Spectrum Analyzer according to the following procedure. An AQ6370 screen is used for this example.

Procedure

Selecting the Communication Interface

- Press SYSTEM. The system setup menu appears.
- Press the MORE1/4 soft key. The communication interface setup menu appears.
- Press the REMOTE INTERFACE soft key. A menu for selecting the interface appears.
- 4. Press the ETHERNET soft key to set the communication interface to Ethernet.



Setting the TCP/IP Parameters

- Press the NETWORK SETTING soft key. The Ethernet setup menu appears.
- 6. Press the TCP/IP SETTING soft key.

The TCP/IP setup menu appears.



- 7. Press the < and > soft keys to select AUTO (DHCP) or MANUAL.
- 8. Press the SELECT soft key.
 - A check is placed by the selected item.
 - If you select AUTO, proceed to step 10.
- 9. If you select MANUAL, set the IP address, subnet mask, and default gateway. Press the <, >, ^, and v soft keys to select the input position, and press ENTER.
- 10. Enter the value using the rotary knob or arrow keys, and press ENTER.
- **11.** After entering all the settings, press the **DONE** soft key.

Setting the Remote Port Number

- **12.** Press the **REMOTE PORT NO.** soft key. The port number setup screen appears.
- 13. Enter the port number using the rotary knob or arrow keys, and press ENTER.



- Set the remote port number

Check that this is set to AQ6370.

Setting the User Account Information

14. Press the USER ACCOUNT soft key.

The user account information setup screen appears.

15. Set the user name and password used to authenticate the remote connection.



2

Explanation

Setting the TCP/IP Parameters

The IP address of the Optical Spectrum Analyzer must be set correctly to use this function correctly.

If a DHCP server is available on the network to which the Optical Spectrum Analyzer is connected, the IP address is automatically assigned. In this case, set [SYSTEM] <NETWORK SETTING><TCP/IP SETTING> IP ADDRESS SETTING to AUTO. For details on the network to which the Optical Spectrum Analyzer is connected, consult your network administrator.

Setting the REMOTE PORT NO.

Set the port number used to perform remote control from the software program. The default value is 10001.

Connection

The Optical Spectrum Analyzer can connect to a single controller (external PC, etc.). If a connection request is received from a controller when the Optical Spectrum Analyzer is already connected, the current connection is maintained.

Computer Name

The computer name of the Optical Spectrum Analyzer is as follows:

AQ6360: "	'AQ6360 @@@@@@@@@@
AQ6370: "	'AQ6370 @@@@@@@@@@
AQ6370B: "	'6370B @@@@@@@@@"
AQ6370C: "	'6370C @@@@@@@@@@"
AQ6370D: "	'6370D @@@@@@@@@@"
AQ6373: "	'6373 @@@@@@@@@@"
AQ6373B: "	'6373B @@@@@@@@@@"
AQ6374: "	'6374 @@@@@@@@@@"
AQ6375: "	'AQ6375 @@@@@@@@@@"
AQ6375B: "	'6375B @@@@@@@@@"
AQ6376: "	'6376 @@@@@@@@@@"
AQ6377: "	'6377 @@@@@@@@@@"

Where @@@@@@@@@@@@ is the serial number. The serial number is indicated by nine alphanumeric characters on the rear panel of the Optical Spectrum Analyzer. The computer name cannot be changed.

Setting the User Account

The software program performs authentication using the user name and password when making a remote connection to the Optical Spectrum Analyzer.

The software program and the Optical Spectrum Analyzer use the MD5 algorithm (RSA Data Security, Inc. MD5 Message Digest Algorithm) for the password authentication. The default user name is anonymous. When using this user name, the authentication password is not necessary.

3.1 Main Menu

This section explains how to use the main menu of the model-specific software using the AQ6370 Viewer window as an example.

File Menu

Chapter 3

Print

Choose this command to output a hard copy of the screen to a printer connected to the PC. For the operating procedure, see section 3.2, "Making a Hard Copy of the Screen."

Print		<u>?</u> ×
Printer		
Name:	Oki OL870-PS v2013.10(1035P	S) Properties
Status: Type:	Ready Oki OL870-PS v2013.108	
Where: Comment:	IP_10.0.228.42	Print to file
Print range		Copies
• All		Number of copies: 1 📑
C Pages C Select	, , , , , , , , , , , , , , , , , , , ,	123 123
		OK Cancel

Exit

Choose this command to close the program.

Connect Menu

Choose this menu to connect to the Optical Spectrum Analyzer via the Ethernet interface. When connected, the operation mode switches to Remote Control mode.

For the operating procedure, see section 5.1, "Connect and Disconnect Operations."

Connection	
Computer name or IP address:	AQ6370xxxx
Port number:	10001
User name:	anonymous
Password:	
Connect	Cancel

Disconnect Menu

Choose this menu to stop the connection to the Optical Spectrum Analyzer.

When you select the menu, the dialog box shown below appears. Click OK to stop the connection and switch back to Viewer mode.

For the operating procedure, see section 5.1, "Connect and Disconnect Operations."

Disconnection			
Are you su	ire?		
YES	NO		
N			

File Transfer Menu

Choose this menu to transfer the data stored on the Optical Spectrum Analyzer to the PC. For the operating procedure, see section 6.1, "File Transfer Operation."

2			
ownload Upload Delete New Folder	Exit		
AQ6370 File List			
Computer name / IP Address: 192.168.1.100			
Address: INT:¥			Memory: 💽 INT 🛛 🖸
File name	Size	File Type	Date & Time
W0000.CSV	26 KB	CSV File	2006/08/07 15:19:30
🗋 W0000.WV6		WV6 File	2006/07/10 10:55:34
W0001.CSV	26 KB	CSV File	2006/08/07 15:19:54
W0001.WV6	5 KB	WV6 File	2006/07/10 10:55:38
W0002.CSV	26 KB	CSV File	2006/08/07 15:19:58
W0003.CSV	26 KB	CSV File	2006/08/07 15:20:02
°C File List Address: [◯¥AQ63∞_wave¥			
File name	Size	File Type	Date & Time
 060419		File Folder	2006/07/31 22:07:58
17		File Folder	2006/05/11 14:04:27
		File Folder	2006/08/02 18:55:34
19		File Folder	2006/07/26 14:28:58
		Flie Folder	
		File Folder File Folder	2006/08/02 18:50:05
ing no ingaa intEDFA2 intest		File Folder	2006/08/02 18:50:05

Setting Menu

Folder Setting

Choose this command to change the folder (user folder) on the PC for saving the file loaded in Viewer or Remote Control mode or the drive to which the USB storage device is connected. By default, the folder in which this program is stored is the user folder. For the operating procedure, see section 3.3, "Setting the User Folder."

		s¥AQ6370Viewer¥User	
			1.2.1
XT] USB Storage Drive	: D: •		

Display Menu

Trace List (for AQ6360, AQ6374, AQ6375, AQ6375B, AQ6376 and AQ6377)

Lists the optical spectrum analyzer models that measured the waveform data of each trace.



Actual resolution table (for AQ6375, AQ6375B and AQ6376)

Lists the actual resolutions (typical values) determined from the center wavelengths of optical spectrum analyzer models (AQ6375/AQ6375B/AQ6376) for each resolution setting.

*** •	Ctual resoluti -HORIZON SC © nm C		-1	<u>[</u>	Select the unit.
	CENTER: 300	0.000 <u>*</u> nm –			Set the center value (wavelength, frequency, wavenumber)
	Resolution	AQ6375	AQ6375B	AQ6376	
	0.05nm	0.049nm	0.052nm	-)	
	0.1nm	0.075nm	0.082nm	0.102nm	
	0.2nm	0.151nm	0.165nm	0.204nm	Actual resolutions (typical values)
	0.5nm	0.302nm	0.330nm	0.510nm	Actual resolutions (typical values)
	1nm	0.756nm	0.826nm	1.021nm	
	2nm	1.514nm	1.654nm	2.042nm	
		ОК			

Resolution settings

About Menu

Choose this menu to check the software version.

🎬 About AQ6373 Viewer Software 🛛 🔀	
AQ6373 Viewer	— Software version
Version 01.01	
(R01.01)	— Main unit's firmware version corresponding to the analysis
Copyright (C)2009 Yokogawa Electric Corporation	,
This product licence to:	results of this software
This product licence to.	(AQ6376 firmware version in the case of the AQ6375&76
	Viewer)
OK Input License Code	

Note -

- The version Rxx.xx (where xx are numbers) shown when you choose the About menu corresponds to the firmware version. If the Optical Spectrum Analyzer firmware version does not match the version shown here, the analysis results of this program may not match the those of the Optical Spectrum Analyzer.
- The software version shown when you choose SYSTEM > VERSION in Remote Control mode is the Optical Spectrum Analyzer firmware version.

3.2 Making a Hard Copy of the Screen

You can print the displayed screen by carrying out the procedure below.

Procedure

1. On the **File** menu, choose **Print**.

The print setup window opens.		
AQ6370 OPTICAL SPECTRUM AN/ File(F) Connect(Q) Disconnect(D) Fil	∭ Print	<u>? ×</u>
File(E) Connect(Q) Disconnect(D) Fil Print(P) Ctrl+P 370 0PTICAL	Printer	
Exit⊗	Name: Oki OL870-PS v2013.10(1035PS)	▼ Properties
	Status: Ready	
	Type: Oki OL870-PS v2013.108	
	Where: IP_10.0.228.42 Comment:	Print to file
	Commerie	
	Print range	Copies
	© All	Number of copies: 1 🛨
	C Pages from: 0 to: 0	23 23
	C Selection	123 123
	L	
		OK Cancel

2. Set the printer, the number of copies, and the like, and click Print.

3.3 Setting the User Folder

You can change the user folder for saving the file loaded in Viewer or Remote Control mode (or the drive to which the USB storage device is connected) by carrying out the procedure below.

Procedure

1. On the **Setting** menu, choose **User Folder Setting**. The folder setup window opens.



2. Select the folder or drive, and click OK.

3

4.1 Loading Waveform Data and Displaying Waveforms

You can load the waveform data and display the data on the screen by carrying out the procedure below. An AQ6370 Viewer screen is used for this example.

Procedure

Preparing the Waveform Data

- **1.** Copy the waveform data file you want to load from the USB storage medium to the user folder.
 - For the procedure to set the user folder, see section 3.3, "Setting the User folder."
 - For the procedure to save the waveform data to the USB storage medium, see section 8.3, "Saving/Loading Displayed Data" in the *Optical Spectrum Analyzer User's Manual*.
 - To load the waveform data file stored on the USB storage medium, insert the USB storage medium into the PC.

Loading the Waveform Data

 On the task bar, click the Start button, point to Programs > Yokogawa > AQ6370 Viewer (AQ6360 Viewer/AQ6370B Viewer/AQ6370C Viewer/AQ6370D Viewer/ AQ6373 Viewer/AQ6373B Viewer/AQ6374 Viewer/AQ6375&76 Viewer/AQ6377 Viewer) to start the program.

You can also double-click the shortcut icon on the desktop.



4

4.1 Loading Waveform Data and Displaying Waveforms

- 3. Right-click in the program window to show the front panel key menu.
- 4. On the front panel key menu, click FILE.



5. Check that ITEM SELECT is set to TRACE on the soft key menu, and click **READ** on the soft key menu.



If ITEM SELECT is not set to TRACE, click ITEM SELECT to select TRACE.

- 6. Select the file you want to load and click **EXECUTE** on the soft key menu.
 - The files in the user folder are displayed. For the procedure to change the user folder, see section 3.3, "Setting the User Folder."
 - To load the file directly from the USB storage medium, click MEMORY on the soft key menu to select EXT, and carry out this step.

For the procedure to change the drive to which the USB storage medium is connected, see section 3.3, "Setting the User Folder."



Displaying the Waveform

 On the soft key menu, click RETURN. The waveform of the loaded data is displayed as shown below.



4
4.2 Saving the Setup Data

You can save the measurement and display conditions you set in Viewer mode by carrying out the procedure below.

An AQ6370 Viewer screen is used for this example.

Procedure

- 1. Right-click in the program window to show the front panel key menu.
- 2. On the front panel key menu, click FILE.



3. On the soft key menu, click ITEM SELECT.



4. On the soft key menu, click **SETTING**.



5. On the soft key menu, click **WRITE**. The file list is displayed.



6. On the soft key menu, click FILE NAME.



- 7. Type the file name from the keyboard.
 - You can also enter the file name by using the characters and soft keys shown in the character input window. For the input procedure, see section 4.3, "Entering Numerical Values and Strings" in the *Optical Spectrum Analyzer User's Manual*.



8. On the soft key menu, click **DONE**. The file name is entered.



9. On the soft key menu, click **EXECUTE**. The file is saved, and the file name appears in the file list.



4.3 Soft Key Menu in Viewer Mode

Of the soft key menu commands in Viewer mode, those that are not available and those that are available but cannot be executed are summarized in the following tables. Some of the functions may not be listed depending on the model.

For details on the soft key menus, see the user's manual of the relevant optical spectrum analyzer.

Unavailable Menu Commands

Front Panel Key Menu	Soft Key Menu	Functions
SWEEP	AUTO	Sweep
	REPEAT	
	SINGLE	
	STOP	
	SEGMENT MEASURE	
DISPLAY	DISPLAY OFF	Display off
MARKER	MARKER LIST PRINT	Print out the marker value
ANALYSIS	RESULT PRINT	Print out the analysis result
FILE	REMOVE USB STORAGE	Remove the USB storage medium
PROGRAM	PROGRAM EDIT > EDIT > LIST PRINT	Print out the program list
	OUTPUT WINDOW > PRINT OUTPUT WIN	Print out the OUTPUT WINDOW
ADVANCE	DATA LOGGING > START	Executes data logging
SYSTEM (MORE 1/4)	OPTICAL ALIGNMENT	Execute the alignment adjustment of the monochrome meter
	WL CALIBRATION	Wavelength calibration
SYSTEM (MORE 2/4)	REMOTE INTERFACE	Set the communication interface
	GP-IB SETTING	Set the GP-IB interface
	RS-232 SETTING	Set the RS-232 interface
	NETWORK SETTING	Set the Ethernet interface
SYSTEM (MORE 3/4)	SET CLOCK	Set the date/time
SYSTEM (MORE 4/4)	REMOVE USB STORAGE	Remove the USB storage medium
	VERSION	Display the version and update the software
	SYSTEM INFORMATION	Display the system information
	RES BW CALIBRATION	Execute rms resolution calibration
	SHUT DOWN	Shutdown the Optical Spectrum Analyzer
		system
FEED	-	Feed the paper

Menu Commands That Are Available But Cannot Be Executed

Front Panel Key Menu	Soft Key Menu	Functions
PROGRAM		Execute the program (a portion of the commands not executable)

4.4 Commands That Do Not Operate in the Program Function

The program function operates as follows in Viewer mode.

- PROGRAM EDIT
- All commands can be entered and edited.
- PROGRAM EXECUTE

Commands that include hardware operation such as sweep and printer operations cannot be executed. The table below shows the commands that cannot be executed.

	Command	Operation
SWEEP	AUTO	Skip (not processed)
	REPEAT	Update only the measurement condition
	SINGLE	and the screen
	STOP	
	SEGMENT MEASURE	Update only the measurement conditions and the screen
SYSTEM	OPTICAL ALIGNMENT	Skip (not processed)
	SELF WL CALIBRATION	
	EXT WL CALIBRATION ****.***nm	
	EMIS LINE WL CALIBRATION ****.***nm	
	EXT-GAS WL CALIBRATION ****.***nm	
	AUTO OFFSET ###	Change only the ON/OFF of the soft key
	REMOVE USB STORAGE	Skip (not processed)
	RES BW CALIBRATION	
SPECIAL	COPY ON	Skip (not processed)
COMMAND	PRINTER FEED **	
	PRINT ''	
	PRINT @@@@@	
	PRINT @@@@@;	
	PRINT DATA AREA	
	PRINT OUTPUT WINDOW	
	SEND ** ''	
	SEND ** '';@	
	SEND ** '';@;''	
	RESET OPTION	
	SPOLL **;S	
	SEND RS-232 ''	
	SEND RS-232 '';@	
	SEND RS-232 '';@;''	
	SEND LAN @\$,***** ''	
	SEND LAN @\$,***** '';@	
	SEND LAN @\$,***** '';@;''	1
	RECEIVE **;@\$	1
	SENDR RS-232'';@\$	1
	SENDR LAN @\$,*****'';@\$	1
	SET DELIMITER ###	

5.1 Connect and Disconnect Operations

You can connect to the Optical Spectrum Analyzer that you wan to remotely control by carrying out the procedure below.

An AQ6370 Viewer screen is used for this example.

Procedure

Preparations on the Optical Spectrum Analyzer

 Connect the Optical Spectrum Analyzer to an Ethernet network according to the procedure given in section 2.3, "Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer."

Connection

 On the task bar, click the Start button, point to Programs > Yokogawa > AQ6370 Viewer (AQ6360 Viewer/AQ6370B Viewer/AQ6370C Viewer/AQ6370D Viewer/ AQ6373 Viewer/AQ6373B Viewer/AQ6374 Viewer/AQ6375&76 Viewer/AQ6377 Viewer) to start the program.

You can also double-click the shortcut icon on the desktop.



3. Click the **Connect** menu.

The following Connection window opens.



5

- **4.** Type the computer name (or IP address), port number, authentication user name, and authentication password, and click **Connect**.
 - The computer name of the Optical Spectrum Analyzer is as follows:

```
      AQ6360:
      "AQ6360 @@@@@@@@@@"

      AQ6370:
      "AQ6370 @@@@@@@@@"

      AQ6370B:
      "6370B @@@@@@@@@@"

      AQ6370C:
      "6370C @@@@@@@@@@"

      AQ6370D:
      "6370D @@@@@@@@@@"

      AQ63731:
      "6373 @@@@@@@@@@"

      AQ63731:
      "6373 @@@@@@@@@@"

      AQ63731:
      "6373 @@@@@@@@@@"

      AQ63732:
      "6373 @@@@@@@@@@"

      AQ63751:
      "6374 @@@@@@@@@@"

      AQ63755:
      "AQ6375 @@@@@@@@@"

      AQ63758:
      "6375 @@@@@@@@@"

      AQ6376:
      "6376 @@@@@@@@@"

      AQ6377:
      "6377 @@@@@@@@@@"
```

Where @@@@@@@@@@@ is the serial number. The serial number is indicated by nine alphanumeric characters on the rear panel of the Optical Spectrum Analyzer. The computer name cannot be changed.

- For the port number of the Optical Spectrum Analyzer, enter the value you specified when you carried out the procedure given in section 2.3, "Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer." The default value is 10001.
- · The default authentication user name is anonymous.
- If a connection cannot be established, an error message (see section 7.2, "Message and Corrective Actions") is displayed.
- The software program and the Optical Spectrum Analyzer use the MD5 algorithm (RSA Data Security, Inc. MD5 Message Digest Algorithm) for the password authentication.

Disconnection

5. Click the **Disconnect** menu.

The following Disconnection window opens.



6. Click YES.

Viewer mode is activated.

5.2 Precautions to Be Taken When Operating in Remote Control Mode

Front Panel Key Menu Operation in Remote Control Mode

If you operate the front panel key menu in Remote Control mode, the following process is carried out.

Process
Sends a command that corresponds to the soft key.
Sends a command that corresponds to the soft key.
Processed by the program (a command is not sent).
Sends a command that corresponds to the soft key (repeat sweep is
a repetition of the single sweep).
Processed by the program (a command is not sent).
Sends a command that corresponds to the soft key.
Sends a command that corresponds to the soft key. The function is
implemented in the program.
Processed by the program.
Processed by the program (a command is not sent).
Processed by the program (a command is not sent).
Processed by the program (a command is not sent).
Processed by the program.
Processed by the program.
Processed by the program.
Processed by the program (a command is not sent).
Processed by the program (a command is not sent).
Sends a command that corresponds to the soft key. The function is
implemented in the program.
Processed by the program (a command is not sent).
Processed by the program (a command is not sent).
Sends a command that corresponds to the soft key.
Processed by the program (a command is not sent).

Auto Offset Processing

If auto offset is executed on the Optical Spectrum Analyzer while the software program is running, the communication sequence may be disturbed, and proper operation may be hindered.

Therefore, the program turns the auto offset function OFF when a connection to the Optical Spectrum Analyzer is established and instructs the execution of the auto offset every 10 minutes. However, if measurement is in progress at the time the auto offset function is to be executed, the auto offset is executed after the sweep is completed (same as the operation on the Optical Spectrum Analyzer).

Remote Connection and Disconnection

When a remote connection is established, the waveform information, measurement conditions, and analysis parameter settings of the Optical Spectrum Analyzer are loaded into the PC.

When the remote connection is cut, the measurement conditions and analysis parameter settings on the PC are transferred to the Optical Spectrum Analyzer, and the measured waveform on the Optical Spectrum Analyzer is cleared.

Waveform Loading During the Sweep Operation

During the sweep operation, the program retrieves the waveform being measured once a second from the Optical Spectrum Analyzer and displays the waveform.

AQ6360 Front Panel Keys

In Remote Control mode of AQ6360 Viewer, the contents of the AQ6360 main menu are shown as panel keys.

5

5.3 Soft Key Menu in Remote Control Mode

Of the soft key menu commands in remote control mode, those that have the following limitations are summarized in tables. Some of the functions may not be listed depending on the model.

For details on the soft key menus, see the user's manual of the relevant optical spectrum analyzer.

- Unavailable menu commands
- Menu commands that are available but would cause discrepancies with the waveforms on the optical spectrum analyzer.
- Menu commands whose execution progress is not displayed
- Menu commands that are not executed on the optical spectrum analyzer.

Unavailable Menu Commands

Front Panel Key Menu	Soft Key Menu	Functions
MARKER	MARKER LIST PRINT	Print out the marker value
ANALYSIS	RESULT PRINT	Print out the analysis result
FILE	REMOVE USB STORAGE	Remove the USB storage medium
PROGRAM	PROGRAM EXECUTE > EXECUTE	Execute the program
	PROGRAM EDIT > EDIT > LIST PRINT	Print out the program list
	OUTPUT WINDOW	Display out the OUTPUT WINDOW
	EXECUTE 1 to EXECUTE 21	Execute the program that has been registered to the soft key
ADVANCE	DATA LOGGING	Logging function
SYSTEM (MORE 2/4)	REMOTE INTERFACE	Set the communication interface
	GP-IB SETTING	Set the GP-IB interface
	RS-232 SETTING	Set the RS-232 interface
	NETWORK SETTING	Set the Ethernet interface
SYSTEM (MORE 3/4)	SET CLOCK	Set the date/time
SYSTEM (MORE 4/4)	REMOVE USB STORAGE	Remove the USB storage medium
	SHUT DOWN	Shutdown the Optical Spectrum Analyzer system
LEVEL	Main Scale Initialize	Initialize the vertical axis of the main scale (AQ6360)
	Sub Scale Initialize	Initialize the vertical axis of the sub scale (AQ6360)
FEED	-	Feed the paper

Menu Commands That Would Cause Discrepancies with the Waveforms on the Optical Spectrum Analyzer

Front Panel Key Menu	Soft Key Menu	Functions
TRACE (MORE 1/2)	HOLD	Set the active trace to MAX/MIN hold mode
	ROLL AVG	Set the active trace sweep to average mode
DISPLAY	NOISE MASK	Set the noise mask

Menu Commands Whose Execution Progress Is Not Displayed

Front Panel Key Menu	Soft Key Menu	Functions
SWEEP	AUTO	Auto sweep (waveform during auto sweep is not displayed)
SYSTEM (MORE 1/4)	OPTICAL ALIGNMENT	Execute the alignment adjustment of the monochrome meter
	WL CALIBRATION	Wavelength calibration
SYSTEM (MORE 4/4)	RES BW CALIBRATION	Execute rms resolution calibration

Menu Commands That Are Not Executed on the Optical Spectrum Analyzer (Executed only on the PC)

Front Panel Key Menu	Soft Key Menu	Functions
ZOOM	-	Zoom function
MARKER	-	Marker function
PEAK SEARCH	PEAK SEARCH	Execute the peak search
	BOTTOM SEARCH	Execute the bottom search
	SEARCH MODE	Set the search mode
	MULTI SRCH SETTING	Set the multi search
ANALYSIS	-	Analysis function
PROGRAM	PROGRAM EDIT	Edit the program
ADVANCE	TEMPLATE	Go/No Go judgement
SYSTEM (MORE 1/4)	GRID EDITOR	Edit the grid table
	USER KEY DEFINE	Register the user key

6.1 File Transfer Operation

Procedure

Preparations on the Optical Spectrum Analyzer

 Connect the Optical Spectrum Analyzer to an Ethernet network according to the procedure given in section 2.3, "Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer."

You can transfer the file stored on the Optical Spectrum Analyzer to the PC by carrying

out the steps below. An AQ6370 Viewer screen is used for this example.

Connection

 On the task bar, click the Start button, point to Programs > Yokogawa > AQ6370 Viewer (AQ6360 Viewer/AQ6370B Viewer/AQ6370C Viewer/AQ6370D Viewer/ AQ6373 Viewer/AQ6373B Viewer/AQ6374 Viewer/AQ6375&76 Viewer/AQ6377 Viewer) to start the program.

You can also double-click the shortcut icon on the desktop.

3. Click the **FileTransfer** menu.

The following Connection window opens.

If this step is carried out in Remote Control mode, the Connection window does not open, and the File Transfer window opens. In this case, proceed to step 5.

Connection	
Computer name or IP address:	AQ6370xxxx
Port number:	10001
User name:	anonymous
Password:	
Connect	Cancel

- **4.** Type the computer name (or IP address), port number, authentication user name, and authentication password, and click **Connect**.
 - The computer name of the Optical Spectrum Analyzer is as follows:
 - AQ6360:
 "AQ6360 @@@@@@@@@@"

 AQ6370:
 "AQ6370 @@@@@@@@@"

 AQ6370B:
 "6370B @@@@@@@@@"

 AQ6370C:
 "6370C @@@@@@@@@"

 AQ6370D:
 "6370D @@@@@@@@@"

 AQ6373:
 "6373 @@@@@@@@@@"

 AQ6373:
 "6373 @@@@@@@@@@"

 AQ6373:
 "6373 @@@@@@@@@@"

 AQ6373:
 "6374 @@@@@@@@@@"

 AQ6375:
 "AQ6375 @@@@@@@@@"

 AQ6375:
 "6375 @@@@@@@@@@"

 AQ6376:
 "6376 @@@@@@@@@@"

 AQ6377:
 "6377 @@@@@@@@@@"

Where @@@@@@@@@@@@ is the serial number. The serial number is indicated by nine alphanumeric characters on the Optical Spectrum Analyzer rear panel. The computer name cannot be changed.

- For the port number of the Optical Spectrum Analyzer, enter the value you specified when you carried out the procedure given in section 2.3, "Connecting and Setting the Ethernet Interface of the Optical Spectrum Analyzer." The default value is 10001.
- · The default authentication user name is anonymous.
- If a connection cannot be established, an error message (see section 7.2, "Message and Corrective Actions") is displayed.



Download (File Transfer from the Optical Spectrum Analyzer to the PC)

- **5.** Display the transfer destination folder in the PC file list.
- 6. Select the file you want to transfer in the Optical Spectrum Analyzer file list.
- 7. Click Download. You can also choose Download from the File menu.

Note _

- A file whose size exceeds 1 MB cannot be transferred to the PC.
- The maximum number of files and folders that is shown in the list is 4096.

Upload (File Transfer from the PC to the Optical Spectrum Analyzer)

- 5. Display the transfer destination folder in the Optical Spectrum Analyzer file list.
- 6. Select the file you want to transfer in the PC file list.
- 7. Click Upload. You can also choose Upload from the File menu.

Note.

- A file whose size exceeds 1 MB cannot be transferred to the Optical Spectrum Analyzer.
- The maximum number of files and folders that is shown in the list is 4096.

Deleting Files and Folders

- **5.** Select the file or folder you want delete in the Optical Spectrum Analyzer file list or the PC file list.
- 6. Click **Delete**. You can also choose Delete from the File menu.

Note.

Files and folders that are set to read-only cannot be deleted.

Creating a Folder

- **5.** Select the destination in which the folder is to be created in the Optical Spectrum Analyzer file list or the PC file list.
- Click New Folder. You can also choose New Folder from the File menu.
- 7. In the folder name input dialog box, type the name of the folder you want to create, and click **OK**.

7.1 Start the Common Software

Start the common software to display the navigation menu.

Procedure

1. On the Windows start menu, click Programs > Yokogawa > OSA Viewer to start the software.

You can also start the software by double-clicking its shortcut icon on the desktop. A navigation menu appears.



2. Click the button corresponding to the function you want to use.

The window for the selected function appears.

Explanation

Functions

Viewer Software

You can select which viewer to start depending on the data you want to view. You can run the viewer simultaneously with the remote control software. For details, see section 7.2, "Operations on the Viewer Software (Common software).

Remote Control Software

You can remotely control an optical spectrum analyzer connected to the network. You can run the remote control software simultaneously with the viewer software. For details, see section 7.2, "Operations on the Remote Control Software (Common software).

File Transfer Software

Files saved in an optical spectrum analyzer can be transferred to the PC. The function is the same as with the model-specific software. For details, see chapter 6, "File Transfer Operation."

Viewing the Data Save Destination Folder

You can open the user folder set with the model-specific software. You can check the data save destination of the viewer software and the like. For details on setting the user folder, see section 3.3, "Setting the User Folder."

Viewing the User's Manual

You can view a PDF file of the user's manual.

Viewing the Version and License Information of the Software

You can view the version and license information of this software.

7

7.2 Operations on the Viewer Software (Common software)

The function is the same as with the model-specific viewer software. See also chapter 4.

Procedure

Preparing Waveform Data

Copy a waveform data file that you want to load from a USB storage device to the folder specified as the user folder according to the procedure in section 4.1, "Loading Waveform Data and Displaying Waveforms."

Loading Waveform Data

1. On the navigation window, click a Viewer button.

The viewer window appears.

Start the appropriate viewer for the data you want to view.

Button name	Supported models
AQ6360 Series:	AQ6360
AQ6370 Series:	AQ6370D, AQ6370C
AQ6373 Series:	AQ6373, AQ6373B
AQ6374 Series:	AQ6374
AQ6375 Series:	AQ6375, AQ6375B
AQ6376 Series:	AQ6376
AQ6377 Series:	AQ6377

2. Load the waveform data according to the procedure in section 4.1, "Loading Waveform Data and Displaying Waveforms."

You can also use Open File of the main menu to load the waveform data.

The viewer software operation is the same as the model-specific viewer software. See chapter 4.

Main Menu

You can use the following functions from the main menu.

🖬 AQ6373(Trial version 30days remaining) — 🗆 🗙			
🖬 Load Trace 📕 Save Screen 🖶 Print 🏟 Folder Setting 🏢 Trace List 🕕 Actual Resolution Table			
Actual Resolution Table			
Lists the actual resolutions (typical values) for the			
center wavelengths of the AQ6375/AQ6375B/			
AQ6376/AQ6360 for each resolution setting.			
See section 3.1, "Main Menu."			
Trace List			
Lists the optical spectrum analyzer models that			
measured the waveform data of each trace.			
See section 3.1, "Main Menu.			
Folder Setting			
Set the folder displayed on the Viewer's File menu.			
Print			
Prints the displayed screen. See section 3.2, "Making a Hard Copy			
of the Screen."			
Save Screen			
Saves the displayed screen as an image file (BMP, JPEG, or PNG format).			
Load Trace			

Select the waveform data to load.

Load Trace

The user folder set in section 3.3 opens. Select the waveform data to show in the viewer software, and click Open.

Save Screen

The user folder set in section 3.3 opens. Set the file name and data format of the image file, and click Save.

Print

A print setup window appears. Set the target printer, number of copies, and so on, and click Print.

Folder Setting

Set the folder displayed on the Viewer's File menu.

You can specify a PC's internal storage (INT) and external storage connected to the PC (EXT).

Trace List (for the AQ6360/AQ6374/AQ6375/AQ6375B/AQ6376/AQ6377)

Lists the optical spectrum analyzer models that measured the waveform data of each trace.

Actual Resolution Table (for the AQ6375/AQ6375B/AQ6376)

Lists the actual resolutions (typical values) for the center wavelengths of the optical spectrum analyzer for each resolution setting.

7.3 Operations on the Remote Control Software (Common software)

The function is the same as with the model-specific remote control software. See also chapter 5.

Procedure

1. On the navigation window, click the **Remote Control** button.

A connection setup window appears.

2. As described in section 5.1, "Connect and Disconnect Operations," enter the computer name or IP address, port number, authentication user name, and password. Then click **Connect**.

A Remote Control window appears.

The operation of the Remote Control Software is the same as the model-specific Remote Control Software. See chapter 5.

Main Menu

You can use the following functions from the main menu.

E	Remote Monitor(192.168.1.12:20001,915810516) — 🗆 🗙	
G	Load Trace 🖥 Save Trace 🔍 Large 🔳 Medium 🔍 Small	
C	YOKOGAWA 🔶 A06370D #200 Marchan Marchan IP Address : Serial Number : Serial Number :	
	Serial number and IP address of the connected device	е
	Large, Medium, Small	
	Sets the size of the displayed window	
	Save Trace	
	Saves an image data of the measurement data or window shown on the optical	
	spectrum analyzer to the PC	

Load Trace

Loads the selected measurement data or setup data into the optical spectrum analyzer

Load Trace

The user folder set in section 3.3 opens. Select the data to load into the connected device, and click Open.

Save Trace

The user folder set in section 3.3 opens.

Set the file name and the data to save, and click Save. To save image data, you can set the data format to BMP or TIFF through remote control.

You can also set the color to Black & White, Color, or Preset Color.

Large/Midium/Small

Set the size of the Remote Control Software window. Large: Sets the size to 150% of the medium setting Small: Sets the size to 50% of the medium setting

Explanation

The function is the same as with the model-specific remote control software. You can also display the viewer simultaneously.

You can save waveform data using Remote Control and view it with the Viewer.

Load Trace

You can load the following types of data.

- Trace (*.WXA, *.WX7 to *.WX9): Waveform data measured with the same model as the connected device
- Trace (*.CSV): Measurement data measured with the connected device (CSV format)
- Setting file (*.STA, *.ST7 to *.ST9)

The extension of the data that can be loaded varies depending on the connected model.

Model	Waveform Data	CSV	Setup data
AQ6360	*.WVA	*.CSV	*.STA
AQ6370C/AQ6370D	*.WV8	*.CSV	*.ST8
AQ6373/AQ6373B/AQ6374	*.WV9	*.CSV	*.ST9
AQ6375/AQ6375B/AQ6376/AQ6377	*.WV7	*.CSV	*.ST7

Save Trace

You can save the following types of data.

- Trace (*.WXA, *.WX7 to *.WX9): Waveform data measured with the connected device
- Trace (*.CSV): Measurement data measured with the connected device (CSV format)
- All Trace (*.CSV): All trace measurement data measured with the connected device (CSV format)
- Image: Image data of the waveform screen of the connected device (*.BMP, *.TIFF)

8.1 Troubleshooting

If servicing is necessary, or if the program is not operating correctly after performing the corrective actions described below, contact your nearest YOKOGAWA dealer.

Description	Probable Cause/Corrective Action	Reference Section
Cannot connect to the Optical Spectrum Analyzer (Error message: Connection Error)	 Check that the computer name or IP address that you entered in the Connection window matches with that specified on the Optical Spectrum Analyzer. Check that the Optical Spectrum Analyzer and the PC are connected to the network correctly using LAN cables. Change the Optical Spectrum Analyze's remote interface setting to GP- 	2.3
	IB, then set it back again to ETHERNET.	
Cannot connect to the Optical Spectrum Analyzer (Error message: User account data	Check that the computer name or IP address that you entered in the Connection window matches with that specified on the Optical Spectrum Analyzer.	2.3
mismatch.)		5.2
Cannot connect to the AQ6370 (Error message: AQ6370 Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6370 is set to "ETHERNET." Check that the AQ6370 firmware version is R02.01 or later.You can check the version by pressing the SYSTEM key followed by the VERSION soft key. Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6370B (Error message: AQ6370B Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6370B is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP- IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6370C (Error message: AQ6370C Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6370C is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6370D (Error message: AQ6370D Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6370D is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6373 (Error message: AQ6373 Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6373 is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6373B (Error message: AQ6373B Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6373 is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
Cannot connect to the AQ6374, AQ6375, AQ6375B, or AQ6376 AQ6377, (Error message: OSA Remote interface setting or Firmware Version Error.)	 Check that the remote interface of the AQ6374, AQ6375, AQ6375B, AQ6376 or AQ6377 is set to "ETHERNET." Change the Optical Spectrum Analyze's remote interface setting to GP-IB, then set it back again to ETHERNET. 	2.3
The license authentication window appears when starting the program	Execute license authentication. The license code needed for the license authentication is indicated on the accompanying CD case.	2.1
The analysis result of the program does not match the analysis result on the Optical Spectrum Analyzer.	The version Rxx.xx (where xx are numbers) shown when you choose the About menu corresponds to the Optical Spectrum Analyzer firmware version. Check that this firmware version and the Optical Spectrum Analyzer firmware version match.	3.1
	 Confirm that the waveform analysis was conducted on the Viewer that corresponds to the optical spectrum analyzer that measured the waveform data. Otherwise the following analysis results cannot be guaranteed (including waveform data from the AQ6317 or AQ6319). Power analysis, LED analysis, WDM analysis, OSNR (WDM) analysis, EDFA-NF analysis, dBm/nm mode, and dBm/THz mode 	-

8.1 Troubleshooting

Description	Probable Cause/Corrective Action	Reference Section
The waveform measured in Remote Control mode and displayed on the PC does not match the waveform displayed on the Optical Spectrum Analyzer.	 Waveforms do not match when the noise mask function is used or when using the following trace settings. ROLL AVG/MAX HOLD/MIN HOLD/CALC Change the Optical Spectrum Analyze's remote interface setting to GP- IB, then set it back again to ETHERNET. 	_
Large files cannot be transferred in File Transfer mode.	The maximum file size per file that can be transferred in File Transfer mode is 1 MB.	_

8.2 Messages and Corrective Actions

Error Messages

The following error messages may appear while running the software program.

Error Message	Description
	Communication with the Optical Spectrum Analyzer failed due to any of the following reasons.
Connection Error	• An error occurred in the connection to the Optical Spectrum Analyzer (this error occurs also if the Optical Spectrum Analyzer is already connecting from a different software program).
	An error occurred while sending data to the Optical Spectrum Analyzer.
	A timeout occurred while receiving data from the Optical Spectrum Analyzer.
Connection Error. AQ6370 Remote interface	 The remote interface of the target AQ6370 is not set to ETHERNET.
setting or Firmware Version Error.	• The firmware version of the target AQ6370 does not support this software program. The AQ6370 firmware version must be R02.01 or later.
Connection Error.	
AQ6370B Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6370B is not set to ETHERNET.
Connection Error. AQ6370C Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6370C is not set to ETHERNET.
Connection Error. AQ6370D Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6370D is not set to ETHERNET.
Connection Error. AQ6373 Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6373 is not set to ETHERNET.
Connection Error. AQ6373B Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6373B is not set to ETHERNET.
Connection Error. OSA Remote interface setting or Firmware Version Error.	The remote interface of the target AQ6374/AQ6375/AQ6375B/ AQ6376/AQ6377 is not set to ETHERNET.
Connection Failed. User account data mismatch.	 The account information used to connect to the Optical Spectrum Analyzer does not match with the information set on the Optical Spectrum Analyzer. The Optical Spectrum Analyzer could not respond because auto offset was in progress when the software program tried to connect to the Optical Spectrum Analyzer.

Warning Messages during Remote Control

Because the software program retrieves error information from the Optical Spectrum Analyzer during remote control, if an error is detected on the Optical Spectrum Analyzer, the warning message for the error is also displayed on the software program. For the cause and corrective action for the warning messages that are displayed while running the program, see section 10.10, "Warning Display Function" in the *Optical Spectrum Analyzer User's Manual.*

Index

Δ

Α	Page
About	
Actual Resolution Table	
AUT ANA	
AUT CTR	1-9
AUT OFS	
auto offset processing	
AUT REF	1-9
AUT SCL	1-9
AUT SRC	
C	Page

С	Page
commands that do not operate in the program function	
computer name2-8, 4	5-2, 6-1
Connect	3-1, 5-1
connection	5-1, 6-1
COR SIZ	

D

D	Page
Delete Disconnect disconnection Display Menu DONE Download.	

<u>E</u>	Page
error messages	8-3
Ethernet connection	
Ethernet settings	
EXECUTE	4-3, 4-7
EXECUTE	2-2, 3-1

<u>F</u>	Page
File FILE FILE NAME FileTransfer file transfer mode folder	3-1 4-2, 4-4 4-5 3-2, 6-1 1-3 6-3
Folder Setting front panel key menu	

<u>I</u>	Page
installation ITEM SELECT	
К	Page
keyboard	
L	Page

i age
Page
1-1, 4-8
Page

O Page
operation mode
P Page
package, contents ofiv parameter value
R Page
READ
S Page
Save Screen 7-3 Save Trace 7-4 screen, hard copy 3-4 Setting 3-2 SETTING 4-5 setup data, saving of 4-4 SGL 1-9 SMO OTH 1-9 soft key menu 4-8 soft key menu shapes and functions 2-5 SPT 1-9 SRC 1-2 1-9 SRC ZOM 1-9 start 2-1 status bar 1-10 sweep operation, waveform loading during 5-3 SWP 1-2 1-9 SWP 1-2 1-9 SWP 1-2 1-9 SWP INT 1-9 system requirements v V Page TCP/IP 2-7 TCP/IP setup 2-8 TLS SYC 1-9 Trace List 3-3, 7-3
U Page
Update 1-7 Upload 6-2 User Account 2-7, 2-8 User Folder Setting 3-5
V Page
VAC WL
W Page
warning messages